Author's response to reviews

Title: Urban air pollution and emergency room admissions for respiratory symptoms: a case-crossover study in Palermo, Italy.

Authors:

Fabio Tramuto (fabio.tramuto@unipa.it)
Rosanna Cusimano (rcusimano@unipa.it)
Giuseppe Cerame (cerameg@libero.it)
Marcello Vultaggio (vultaggio.marcello@amianet.it)
Giuseppe Calamusa (gcalamusa@unipa.it)
Carmelo M Maida (c.maida@unipa.it)
Francesco Vitale (francesco.vitale@unipa.it)

Version: 5 Date: 13 April 2011

Author's response to reviews:

The text and the additional files have been revised according to the editorial suggestions.